Author's response to reviews

Title: Efficacy of citric acid-based denture cleanser on Candida albicans biofilm formed on poly(methyl methacrylate): effects on residual biofilm and recolonization process

Authors:

Fernanda Faot (fernanda.faot@gmail.com)
Yuri W Cavalcanti (yuri.wanderley@yahoo.com.br)
Martinna M Bertolini (martinnaberloni@gmail.com)
Luciana R Pinto (lucianaderezende@yahoo.com.br)
Wander José da Silva (wanderjose@fop.unicamp.br)
Altair A Del Bel Cury (altcury@fop.unicamp.br)

Version: 2
Date: 7 April 2014

Author's response to reviews:

Line and page number were provided.